Author's response to reviews

**Title:** Access to Dental Public Services by Disabled Persons

**Authors:**

Lyana L Rocha (lyanafurtado3@hotmail.com)
Maria V L Saintrain (mvlsaintrain@yahoo.com.br)
Anya P G F Vieira-Meyer (anyavieira10@gmail.com)

**Version:** 2  
**Date:** 23 February 2015

**Author's response to reviews:** see over
Letter to the editor:

It is with great pleasure that we resubmit this manuscript to your journal. As requested, we are sending (below) a point by point answer to the reviewers. Do not hesitate in contacting us if you need any further information.

Best regards,
Anyia Vieira, Maria Vieira and Lyana Rocha

Reviewer's report
Title: Access to Dental Public Services by Disabled Persons
Version: 1
Date: 2 February 2015
Reviewer: Soraya Leal

Reviewer's report: Dear authors,
The paper had improved in its quality and has merits to be published.
Level of interest: An article whose findings are important to those with closely related research interests
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.
Declaration of competing interests: I declare that I have no competing interests

We thank the reviewer for her comments.

Editorial request:
1. We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bme1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

We sent the manuscript to be edit at Edanz. The result this editing process can be visualized in the new manuscript.
2. Please change 'introduction' to 'Background'.

*Change performed and marked in red in the text.*

3. Please include 'Study limitations' in the Discussion section.

*Inclusion performed and marked in red in the text.*

4. Please include list of Abbreviations.

*Inclusion performed and marked in red in the text.*